
Wednesday and Thursday:  DVM○

On-call:  

Controls experts changed out the M:IP107 power supply○

 <stacking rate> = 26.7 mA/hr
 <production> = 25.0 pbar/Mp
 <POT> = 8.43 Tp

○ Stacking Numbers

 <stacking rate> = 28.0 mA/hr
 <production>=25.8 pbar/Mp
 <POT>=8.44 Tp

○ Overnight (20:00 to 02:00) not including large stack

Stacking

Had a problem with the stacktail not gaiting off for transfers.  Settings peeker showed that the 
commands were going out at the correct time, but both the pin switch and medium level amp 
stayed on.  No errors were reported by the sequencer.    Rebooted PBCOOL front end.  All 
seemed better.  http://www-bd.fnal.gov/cgi-mach/machlog.pl?nb=pbar10&action=view&page=
268&anchor=093248&hilite=09:32:48-

○

○ This is similar to a problem on the previous  day, where the sequencer didn't see the $94 after a 
set of transfers and did not turn the stacktail back on.

○ Unstacked 446E10 in 57 transfers in 19 sets.
○ Stacking efficiency = 96.8%

Transfers

○ Check on AP1 vacuum.  
○ Cooldown 1.25 hours?
○ Turbo need an hour
○ Leak check 4 hours.

PreVault Access

Pbar
ID Reques

tor
Title Locatio

n
Type

12509 Gollwit
zer, 
Keith

Inspection of PreVault

Inspection of Pbar tunnel; search for water leaks and ground 
water etc. 

PreVaul
t 
enclosu
re

Misc

12500 Gollwit
zer, 
Keith

Inspection of Transport

Inspection of Pbar tunnel; search for water leaks and ground 
water etc. 

Transp
ort 
enclosu
re

Misc

12499 Gollwit
zer, 
Keith

Inspection of PreVault/Target

Inspection of Pbar tunnel; search for water leaks and ground 
water etc. 

PreVaul
t/Targe
t 
enclosu
re

Misc
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12497 Drendel
, Brian

Leak Check M:IP107

M:IP701 is tripping off. It looks like we have a vacuum leak in 
the vicinity of EB6. It would be helpful to complete a leak check 
of the area. We would like to line this up with other naturally 
occurring downtime. 

PreVaul
t 
Enclosu
re

Vacuu
m

12491 Sprosty
, Susan

ODH EXH FANS

QUARTERLY MAINT.PM.FOR ODH EXH FANS FESS ASSET 
#AP183,AP184,AP185,AP186,AP187 

PBAR 
TUNNE
L

FESS / 
Utilities

12474 Drendel
, Brian

LCW Manifold Leak on A1B3

A1B3 has an LCW leak on the LCW manifold. The leak can be 
seen if you look at the manifold from the downstream end of 
the magnet. Under the 3" in the center of the manifold, there 
is a very slow LCW weaper coming from the threads on that 
connection. This is just above the small orange hose connector 
labled "9."

Pbar 
Rings

Water

12471 Drendel
, Brian

D1Q5 LCW hose replacement

D1Q5 has a slow dripper on the fitting for the orange hose that 
goes to the LCW return header. 

Pbar 
Rings

Water

12470 Gollwit
zer, 
Keith

Inspection of Pbar Rings

Inspection of Pbar tunnel; search for water leaks and ground 
water etc. 

Rings 
enclosu
re

Misc

12469 Drendel
, Brian

Rollaround Console Doesnt Work

Stochastic Cooling guys report that the Thinwire Ethernet 
rollaround cart console was not working when they tried to use 
it in the tunnel in the A30 region. The Thinnet repeater in the 
AP30 service building was power cycled, which did not fix the 
problem. We need the Network Guys to look at the Thinnet to 
see if it is operational.

Pbar 
Rings

Control
s

12468 Levelin
g, 
Anthon
y

Store 10 mm-7 in coffin

10 mm-7 lens/transformer was removed from service on 8/21 
following development of a transformer ground fault. While 
installing 10mm-1 on lens module #2, discovered that the 
upstream lens water line needs to be reworked so that the 
module water line can be connected. Drop 10 mm-1 from 
module #2, rework upstream water line, and finish installation 
of 10 mm-1 onto lens module #2.

AP0 
upper 
vault

Target 
Station

12450 Levelin
g, 
Anthon
y

November target air blower check

Perform periodic maintenance on target blower in November 
2010. Last maintenance was on October 4. 

AP0 
Target 
Hall

Target 
Station

12406 Drendel
, Brian

ARF1 calibration

After changing out the Acopian +5V power supply in the ARF1 
LLRF, the output of the ARF1 amplitude curve is saturated. One 

AP50 Low 
Level 
RF
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LLRF, the output of the ARF1 amplitude curve is saturated. One 
hour of no stacking time is needed to recalibrate the output. 

12378 Drendel
, Brian

D:EKIK Module #1 Power Supplies

D:EKIK module #1 is starting to show some timing drift. During 
a shutdown of a day or longer, it would be nice to open up the 
oil-filled thyratron tank and check the brick power supplies. 

AP10 
Service 
Buildin
g

Power 
Supply

12345 Vander 
Meulen
, David

D:V4TW01 Tripps

D:V4TW01 is tripping on reverse power. A bad cable was 
replaced in the tunnel, but the problem is in the tank, which 
means we have to let up vacuum. The job including vacuum 
pump down is a minimum of 6 hours. 

D30 
Tunnel

Stochas
tic 
Cooling

12253 Levelin
g, 
Anthon
y

Install new shielding blocks

New concrete shielding blocks are to be installed for the 
JASMIN experiment. Install shield blocks at some opportunistic 
access when the shield blocks become available. Estimated 
arrival is now late August/early September. 

AP0 
upper 
vault

Target 
Station

12240 Sievert, 
Ken

Connect BV500H to A50 CIA Crate

Terminate cable and connector on valve, modify crate to 
accept valve status. 

A50 Vacuu
m

9310 Obrycki
, Mark

Sled drawings/ panel schedules

Inspect breaker panels in tunnel 

AP-10 
tunnel

ES&H / 
Interloc
ks

Total 
Reques
ts: 17

 Pbars stacked: 582.62 E10
 Time stacking: 22.82 Hr
 Average stacking rate: 25.53 E10/Hr

○ Stacking

 Number of pulses while in stacking mode: 28177
 Number of pulses with beam: 27303
 Fraction of up pulses was: 96.90%

○ Uptime

 Corrected time stacking: 22.11 Hr
 Possible average stacking rate: 26.35 E10/Hr
 Could have stacked: 601.27 E10/Hr

○ The uptime's effect on the stacking numbers

 Pbars sent to the Recycler: 467.54 E10
 Number of transfers : 60
 Number of transfer sets: 20

○ Recycler Transfers

The Numbers
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 Number of transfer sets: 20
 Average Number of transfer per set: 3.00
 Time taken to shoot including reverse proton tuneup: 00.21 Hr
 Transfer efficiency: 97.26%

 Average POT : 8.43 E12
 Average production: 25.32 pbars/E6 protons

○ Other Info

The Plots
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